Author’s response to reviews

Title: Awareness, understanding and use of sodium information labelled on pre-packaged food in Beijing: a cross-sectional study

Authors:
Yao He (heyao_pku@163.com)
Liping Huang (hliping@georgeinstitute.org.cn)
Sijin Yan (yansijin09@163.com)
Yuan Li (yli@georgeinstitute.org.cn)
Lixin Lu (lulixin518@sina.com)
Hongbo Wang (382080864@qq.com)
Wenyi Niu (13693519613@163.com)
Puhong Zhang (zpuhong@georgeinstitute.org.cn)

Version: 3 Date: 24 Mar 2018

Author’s response to reviews:

Dear Editor, First of all, I would like to express my heartfelt thanks to your suggestions and editing of our manuscript. Changes have been made as followed: Background-line 4: we added the word "to", so the sentence now reads as [we use “sodium” to mean “salt”]; Methods-line 17: we have deleted the "RELATIVELY / AND" 'Consent to publish' section: We removed the information currently included in 'Consent to publish' and replace it with “Not applicable”; Authors' contributions' section, we added the specific contributions of Yuan Li. It reads as [The manuscript was drafted by YH and critically revised by LH, YL and other authors] Thank you very much for your work concerning my paper. Wish you all the best! Sincerely yours, Puhong Zhang.